Author’s response to reviews

Title: Standardized Herbal Extract PM014 Alleviates Fine Dust-Induced Lung inflammation in Mice

Authors:

Ye-Seul Lee (jparadise.lys@gmail.com)
Daean Min (wndqr1456@gmail.com)
Seon-Young Park (psys12@naver.com)
Junyoung Lee (gappaman2@nate.com)
Hyunsu Bae (hbae@khu.ac.kr)

Version: 6 Date: 19 Aug 2020

Author’s response to reviews:

– We note that the current submission contains some textual overlap with other previously published works, in particular:

https://ir.ymlib.yonsei.ac.kr/handle/22282913/162228

Also from

https://www.nature.com/articles/s41598-018-35320-8

While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work.

If there is overlap in the Methods section, please ensure to summarize the methods and cite the source.

Please re-phrase these sections to minimise overlap.

– Please ensure that all contributing authors are also listed on our system. (Junyoung Lee)

– At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
- We appreciate the Editor's comment. As the Editor suggested, we checked the manuscript thoroughly using a plagiarism detection program and revised the manuscript accordingly. We hope this sixth revision is satisfactory to your journal's standards.